

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Lawson, Nicholas

Working Paper

Discrimination and the fiscal benefits of immigration

Research Group on Human Capital - Working Paper Series, No. 24-01

Provided in Cooperation with:

University of Quebec in Montreal, School of Management Sciences (ESG UQAM), Research Group on Human Capital (GRCH)

Suggested Citation: Lawson, Nicholas (2024): Discrimination and the fiscal benefits of immigration, Research Group on Human Capital - Working Paper Series, No. 24-01, Université du Québec à Montréal, École des sciences de la gestion (ESG UQAM), Groupe de recherche sur le capital humain (GRCH), Montréal

This Version is available at: https://hdl.handle.net/10419/299328

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Research Group on Human Capital Working Paper Series

Discrimination and the Fiscal Benefits of Immigration

Working Paper No. 24-01

Nicholas Lawson

January 2024



https://grch.esg.uqam.ca/en/working-papers-series/

Discrimination and the Fiscal Benefits of Immigration

Nicholas Lawson* Université du Québec à Montréal

January 29, 2024

Abstract

In recent decades, there has been a lengthy debate about the fiscal costs or benefits of immigration, and much of the literature has found fiscal impacts that are close to zero. However, these studies have ignored the possibility that immigrants may be victims of wage discrimination in the labour market, despite evidence of such discrimination in various countries. In the presence of such discrimination, existing estimates of the fiscal impact of immigration will be biased: if immigrants are paid less than their marginal products, then someone else is receiving that income – mostly likely the firm's owners or other workers – and paying taxes on it, and that fiscal benefit is ignored by a model that disregards discrimination. In this paper, I evaluate the quantitative importance of this mechanism, by calibrating a search-and-matching model to Canadian data and simulating the fiscal impact of increases in immigration. When the model and calibration omits wage discrimination against immigrants, the average fiscal impact of immigration is negative, but it becomes positive if discrimination explains the wage gaps between natives and immigrant workers: at an economy-wide level, an annual fiscal cost of about \$3 billion in the absence of discrimination becomes a fiscal benefit of about \$4 billion in the presence of discrimination. My results indicate that wage discrimination against immigrants could significantly affect our estimates of the fiscal impact of immigration.

Keywords: discrimination, immigration, fiscal benefits

JEL Codes: H27, J61, J79

^{*}lawson.nicholas@uqam.ca. I would like to thank Christ Kévin Bazié for excellent research assistance, and the Groupe de recherche sur le capital humain at UQAM for funding. I thank Casey Warman for comments and suggestions, as well as participants of the Canadian Public Economics Group 2023 Annual Meeting. Any errors or omissions are the responsibility of the author.

1 Introduction

Over the past few decades, there has been a lengthy debate about the fiscal costs or benefits of immigration, based on the question of whether an immigrant pays more in taxes than they receive in public benefits in their destination country. This literature has included papers that focus on specific countries, such as Bødker et al. (2012) and Hansen et al. (2017) on Denmark, Bratsberg et al. (2014) on Norway, Ruist (2015) on Sweden, National Academies of Sciences, Engineering, and Medicine (2017) and Colas and Sachs (2022) on the United States, and Dustmann and Frattini (2014) on the United Kingdom. Other papers have surveyed the debate more broadly (Rowthorn, 2008; Kerr and Kerr, 2011; Preston, 2014; Vargas-Silva, 2015; Edo et al., 2020; Hennessey and Hagen-Zanker, 2020). While there are a few outliers, much of this literature has found relatively small fiscal effects, whether positive or negative, leading to the general conclusion that immigration's fiscal impact is probably close to zero in many countries.

However, these studies have ignored the possibility that immigrants may be victims of discrimination in the labour market. A growing literature provides evidence that immigrants are indeed subjected to discrimination, including direct estimates of wage discrimination in Germany in Bartolucci (2014) and Hirsch and Jahn (2015) and in Belgium in Kampelmann and Rycx (2016) and Fays et al. (2021), as well as more indirect evidence from studies using fake CVs such as Oreopoulos (2011) for Canada and Busetta et al. (2018) for Italy.⁴ To the extent that such discrimination is based on racial and ethnic differences from the majority

¹This question is related to but distinct from the question of the broader economic benefits or costs of immigration, which has also been the focus of a large literature, including Borjas (1995), Boubtane et al. (2014), Battisti et al. (2018), and a number of papers that also consider fiscal impacts (Kerr and Kerr, 2011; National Academies of Sciences, Engineering, and Medicine, 2017; Edo et al., 2020).

²Additionally, some papers have focussed on specific types of immigrants, such as Ruist (2015), who focusses on refugees, or Bødker et al. (2012), who consider high-skilled economic immigrants.

³This includes an interesting recent study by Yao et al. (2022) who used the famous Mariel Boatlift as a plausibly exogenous shock to the fiscal environment in Miami, and find no effect at the municipal level.

⁴An ongoing debate in the literature considers the extent to which immigrant wage gaps may be explained by productive characteristics such as lower host-country language ability (Ferrer et al., 2006; Warman et al., 2015) or lower-quality education (Fortin et al., 2016; Li and Sweetman, 2014). I do not take a position on this, to the extent that I present two cases, with and without discrimination, and demonstrate the impact that discrimination can have if it is present. However, it is important to note that lower returns to education or other skills does not rule out discrimination, as pointed out by Esses et al. (2014): "the ambiguity of immigrants' foreign-acquired skills and personal characteristics may provide a cover for the expression of bias toward immigrants who are religious and ethnic minorities."

population in destination countries – as suggested by Oreopoulos (2011), who finds evidence of unconscious bias against individuals with non-English names – this problem may become even more extensive in the future.

The reason why discrimination against immigrants is important when estimating the fiscal impact of immigration is simple: if immigrants are paid lower wages than comparably productive natives – and thus immigrants are paid less than their marginal products – then someone else is receiving that money as part of their income. That "someone else" could be the firm owner or shareholders, or possibly other workers who are able to negotiate larger wages out of the firm's "discrimination surplus";⁵ but in any case, the recipient will presumably pay taxes on the additional income, which is a fiscal benefit that we should account for when considering the fiscal impact of immigration. In other words, discrimination may be a cost from the immigrant's point of view, but it is a benefit to someone else, and eventually to the government in the form of additional tax revenues. An analysis of the fiscal impact of immigration that does not account for such an effect is likely to be biased.

In this paper, I evaluate the quantitative importance of this mechanism. I present a search-and-matching model based on Battisti et al. (2018), which features 4 types of workers: high-skilled and low-skilled workers (those with and without university degrees) are each subdivided into natives and immigrants. I consider two versions of the model: one in which wage gaps between natives and immigrants are due to productivity differences, and a second in which discrimination is responsible for the wage gaps (in the form of a lower worker wage bargaining parameter). I then calibrate this model to data from Canada, which has a large and growing immigrant population and accordingly has been studied repeatedly in the literature on fiscal impacts of immigration (Grubel and Grady, 2011; Javdani and Pendakur, 2014; Grady and Grubel, 2014; Zhang et al., 2020; Kapsalis, 2021; Montcho et al., 2022).

Using the calibrated model, I simulate three different scenarios of increased immigration, and I find that, in a baseline model without wage discrimination against immigrants, the average fiscal impact of immigration is negative, though high-skilled immigrant workers do produce fiscal gains. When I assume instead that discrimination is the cause of immigrant

⁵This logic implies that it is also possible that papers that show positive effects of immigration on productivity, such as Peri (2012) and Mitaritonna et al. (2017), are partly picking up the spillover effects from discrimination benefitting the firm or the other workers.

wage gaps, the results change significantly: the fiscal contribution of both low- and high-skilled immigrants become more positive, and the overall fiscal impact becomes positive. At an economy-wide level, my estimates imply an annual fiscal cost of about \$3 billion in the absence of discrimination, which becomes a fiscal benefit of about \$4 billion in the presence of discrimination. In other words, accounting for discrimination in my model adds about \$7 billion to the fiscal surplus of Canadian governments.

I also present a series of alternative models and robustness checks: I model part of government expenditure as a pure public good, I apply different tax rates to different worker types, I add taxation on profits or capital to the model, and I use an alternative specification of discrimination. In each case, the numerical results change, but the impact of discrimination on the fiscal environment remains very similar and significant in each case.

My paper contributes to the overall literature on the fiscal impact of immigration, and particularly to the section of the literature that has focussed on Canada: my findings of small negative fiscal impacts in the absence of discrimination are fairly close to most of the existing estimates, with the exception of the much larger negative findings in Grubel and Grady (2011) and Grady and Grubel (2014). Only two papers in the overall literature seem to have addressed the question of discrimination: the first was Javdani and Pendakur (2014), who point out that immigrants might earn less due to discrimination, but claim that this would be a fiscal benefit of removing discrimination rather than recognizing that such a benefit already exists. More recently, Colas and Sachs (2022) mention that, in the monopsonistic labour markets studied by Amior and Manning (2022), immigration implies redistribution between workers and to firms, and that since immigrants are not paid their marginal product, "the economic pie accruing to residents would increase and thereby reinforce the indirect fiscal benefit." However, Colas and Sachs (2022) do not attempt to model or estimate such a fiscal impact of discrimination.

My argument about the importance of an indirect fiscal impact of immigration through discrimination is somewhat similar to that in Clemens (2021), who argues that existing methods to estimate fiscal impacts of immigration ignore the role of capital taxes paid by employers of immigrant labour. I will model such taxes in the extensions of my model in sections 4.3 and 4.4, but my paper is complementary to Clemens (2021) as my main focus

is on discrimination, which is not present in the latter paper's model. Colas and Sachs (2022) also model indirect fiscal effects, through general equilibrium adjustments in wages and employment, which are present in my models.

Finally, it is important to note that there are many different ways that one could construct a model to study the fiscal impacts of immigration. The question of whether to adopt a static or dynamic approach has been addressed numerous times in the literature, including by Vargas-Silva (2015), who also discusses the distinction between simple partial-equilibrium and broader general-equilibrium approaches. Other important questions include how to model the tax system, as well as pensions and social benefits, and how to account for individuals with very high incomes, among many others. Edo et al. (2020) provide a useful summary of the existing research, with a description of the various methods used; to focus on the mechanism that I want to present, I use a relatively simple approach that doesn't account for retirement, childhood, or most government benefits, as I use the static model from Battisti et al. (2018).

The rest of the paper proceeds as follows. Section 2 presents the main model of the paper, while section 3 presents the calibration and the numerical simulations of the effect of discrimination on the estimated fiscal impact of immigration. Section 4 presents a series of alternative models and robustness checks, and section 5 concludes the paper.

2 Main Search and Matching Model

This paper aims to make a simple point: if immigrants face discrimination that leads to them receiving lower wages, the surplus must be received by somebody, and that somebody is likely to pay taxes on it, and this "discrimination surplus" should be taken into account when we estimate the fiscal gains or losses caused by immigration. For such a simple point, a very simple model could be sufficient: for example, if discrimination leads to immigrants' wages being 10% too low – which would be consistent with studies from Germany by Bartolucci (2014) and Hirsch and Jahn (2015) – we could assume that the firm receives that 10% as part of their profits, on which they pay taxes at a rate τ . In that case, the immigrant's fiscal contribution should simply be raised by $0.1\tau w$, if w is the wage of a comparable native worker. If the immigrant's average tax rate t was equal to τ , then we could simply raise

their estimated fiscal contribution in the form of taxes paid by about 11%. Given how many studies have found that the taxes paid and benefits received by immigrants are roughly equal, such an adjustment due to discrimination could easily shift the findings in the literature to being unambiguously positive.

However, such a simple approach is not entirely convincing. First of all, the "discrimination surplus" may be not be entirely received by the firms' owners; native workers may also benefit if they are able to negotiate higher wages to share some of the surplus. Additionally, general equilibrium adjustments to immigration need to be taken into account, in the form of changes to bargained wages and employment rates. And it may be the case that the other people who benefit from discrimination against immigrants – owners and/or other workers – face higher or lower marginal tax rates than the immigrants themselves. To account for all of these factors, we need a more complete model, and that is what the current section of the paper will provide.

Section 4 will consider a variety of model specifications for the sake of robustness, but I start here with a search model based on Battisti et al. (2018), who used their model to estimate the welfare impact of immigration across 20 countries. This model features two skill levels of workers: low-skilled (denoted by L) and high-skilled (denoted by H), which in practice will be identified by whether a worker has completed a university degree. Within each skill type, immigrants and native workers are perfect substitutes and compete in the same skill-specific labour market. Each skill-type of worker produces a separate intermediate good, Y_L and Y_H , which are then combined with perfectly-mobile capital K to produce the single final good Y (for which the price is normalized to 1). As in Battisti et al. (2018), I take the stocks of each type of immigrant to be exogenously given, and thus I only model a single (destination) country. Worker type is denoted by $i \in \{L, H\}$ (skill) and $j \in \{N, I\}$ (native/immigrant). A government provides unemployment insurance (UI) benefits b_{ij} and a lump-sum transfer g, financed by a linear tax t on wage income.

The production technology is unchanged from Battisti et al. (2018):

$$Y = AK^{\alpha}Z^{1-\alpha}, \alpha \in (0,1)$$

$$Z = [xY_L^{\rho} + (1-x)Y_H^{\rho}]^{\frac{1}{\rho}}, \ \rho \in (0,1)$$

$$Y_i = \sum_{j \in N,I} (1-u_{ij})\pi_{ij}Q_{ij}, \ i \in \{L,H\}$$

where α is the output elasticity of capital, A is total factor productivity, ρ determines the elasticity of substitution between intermediate inputs Y_L and Y_H , $x \in (0,1)$ is a relative productivity parameter, Q_{ij} the is total number of individuals of type $\{i,j\}$, u_{ij} is the typespecific unemployment rate, and π_{ij} is labour productivity. Z is a CES composite of the intermediate goods Y_L and Y_H , and the latter are linear functions of each type of labour.⁶

Since capital is freely mobile on world markets, the return to capital r is fixed by its return in those markets. I assume that the stock of capital owned by natives is \bar{K} , which is divided equally among all natives; but the amount of capital K used in production is determined by the first-order condition $r + \delta = \alpha A K^{\alpha-1} Z^{1-\alpha}$, where δ is the depreciation rate. This gives us the condition:

$$(r+\delta)K = \alpha Y. \tag{1}$$

Meanwhile, the intermediate goods are produced under perfect competition, which means that $p_i = \frac{\partial Y}{\partial Y_i}$ for each good:

$$p_L = AK^{\alpha}(1 - \alpha)xY_L^{\rho - 1} \left[xY_L^{\rho} + (1 - x)Y_H^{\rho} \right]^{\frac{1 - \alpha - \rho}{\rho}}$$
 (2)

$$p_H = AK^{\alpha}(1-\alpha)(1-x)Y_H^{\rho-1} \left[xY_L^{\rho} + (1-x)Y_H^{\rho} \right]^{\frac{1-\alpha-\rho}{\rho}}.$$
 (3)

The two skill types have separate labour markets in which matches are formed continuously over time. Intermediate-goods firms pay a cost of c_i per unit of time for an open vacancy, and cannot target their vacancy at immigrants or natives within a skill type. The flow contact rate for each type is given by a Cobb-Douglas matching function over vacancies V_i and a mass $U_i = U_{iN} + U_{iI}$ of unemployed workers:

$$M(U_i, V_i) = \xi U_i^{\varepsilon} V_i^{1-\varepsilon}$$

⁶This sort of CES market structure has been used by a number of other papers that study immigration, including Chassamboulli and Palivos (2014), Chassamboulli and Peri (2020b), and Chassamboulli and Peri (2020a). Of course, such a model implies that, if there are enough high-skill workers, they will actually have lower productivity than low-skill workers, but what matters here is matching the data at and near the baseline equilibrium, where high-skill workers' wages are always higher.

where $\varepsilon \in (0,1)$ is the matching elasticity and ξ is a scale parameter. Labour market tightness is given by $\theta_i \equiv \frac{V_i}{U_i}$, and the rate at which firms fill vacancies is $q(\theta_i) \equiv \frac{M_i}{V_i} = \xi \theta^{-\varepsilon}$, while unemployed workers find jobs at the rate $m(\theta_i) \equiv \frac{M_i}{U_i} = \xi \theta^{1-\varepsilon}$. Existing matches are destroyed at the exogenous rate s_{ij} .

The Bellman equations for firms are identical to Battisti et al. (2018): if we denote as J_i^V the value of an open vacancy and J_{ij}^F as the value of a filled job:

$$rJ_{i}^{V} = q(\theta_{i}) \left[\phi_{iN} J_{iN}^{F} + \phi_{iI} J_{iI}^{F} - J_{i}^{V} \right] - c_{i}$$
$$rJ_{ij}^{F} = \pi_{ij} p_{i} - w_{ij} - s_{ij} \left[J_{ij}^{F} - J_{i}^{V} \right]$$

where $\phi_{ij} = \frac{U_{ij}}{\sum_k U_{ik}}$ is the share of the unemployed of skill level *i* that are of immigration status *j*, and w_{ij} is the worker's wage.⁸ The free-entry condition means that $J_i^V = 0$ in equilibrium.⁹

Meanwhile, the Bellman equations for workers are almost the same as in Battisti et al. (2018): using J_{ij}^E as the value of employment and J_{ij}^U as the value of unemployment:

$$rJ_{ij}^{E} = g + rk_{ij} + (1 - t)w_{ij} - s_{ij} \left[J_{ij}^{E} - J_{ij}^{U} \right]$$

$$rJ_{ij}^{U} = g + rk_{ij} + b_{ij} + m(\theta_i) \left[J_{ij}^{E} - J_{ij}^{U} \right]$$

where k_{ij} is capital per person, which will be an equal share of \bar{K} for natives and zero for immigrants. The one difference with respect to Battisti et al. (2018) is that the latter included h_{ij} as the direct utility from unemployment, where $h_{iN} = 0$ and $h_{iI} < 0$; I drop this for simplicity and to make the distinction between versions of the model with and without discrimination easier to understand.¹⁰

⁷Battisti et al. (2018) divide this separation rate into a baseline separation rate plus a remigration rate for immigrants, but this distinction is without importance for my numerical analysis, so I combine them into a single type-specific rate s_{ij} .

⁸Of course, immigrants' wage trajectories could be different from those of native workers, but I will focus on average wages in the steady-state of this model.

⁹One could question whether firms would be open to hiring both immigrants and native workers; in particular, in the case in which immigrants are subject to wage discrimination, firms may find it particularly profitable to hire immigrants. However, as long as J_{ij}^F is positive for each type of worker, firms will indeed be willing to hire whichever type of worker they meet – and I have verified that this condition is satisfied in all simulations.

 $^{^{10}}h_{iI}$ could be set flexibly to explain the differences in wages between immigrants and natives, without any role for discrimination. Instead, in my model, wage differences between immigrants and natives will be explained by differences in π_{ij} or (in the case of discrimination) by differences in the bargaining power β_{ij} .

Wages are set by Nash bargaining where the worker's bargaining power is $\beta_{ij} \in (0,1)$; unlike in Battisti et al. (2018), I allow the bargaining power to potentially vary across worker type. The worker will then receive β_{ij} percent of the total surplus $J_{ij}^F + J_{ij}^E - J_{ij}^U$, which gives us the following condition:

$$(1 - \beta_{ij}) \left(J_{ij}^E - J_{ij}^U \right) = \beta_{ij} J_{ij}^F. \tag{4}$$

I focus on the steady state of the model, ¹¹ in which flows in and out of unemployment must offset:

$$s_{ij} (Q_{ij} - U_{ij}) = m(\theta_i) U_{ij}$$

which gives us the following results for total employment E_{ij} and unemployment U_{ij} for each type of worker:

$$U_{ij} = \frac{s_{ij}}{s_{ij} + m(\theta_i)} Q_{ij}$$
$$E_{ij} = \frac{m(\theta_i)}{s_{ij} + m(\theta_i)} Q_{ij}.$$

The public sector consists of a government that raises revenues through a linear labour income tax and pays for UI benefits and the lump-sum transfer. I assume that there is no taxation of capital, because domestic capital ownership \bar{K} is fixed by assumption (though utilized capital K adjusts freely in equilibrium) and the rate of return r is fixed by the assumption of a small open economy, so capital income is exogenously fixed; but two alternative models in sections 4.3 and 4.4 will allow for taxes on endogenous capital or on firms. The government budget constraint is:

$$\sum_{i} \sum_{j} b_{ij} U_{ij} + g \sum_{i} \sum_{j} Q_{ij} = t \sum_{i} \sum_{j} w_{ij} E_{ij}$$

$$\tag{5}$$

where, as in Battisti et al. (2018), b_{ij} and g are assumed to be exogenous and t adjusts to satisfy the constraint.

 $^{^{11}\}mathrm{Of}$ course, there could be dynamic effects over time, if employer learning causes workers' productivity to increase and potentially leads to a decrease in discrimination; Warman and Worswick (2015) and Warman et al. (2019) show that immigrant wage gaps in Canada tend to decline with the number of years since immigration, and Green and Worswick (2012) and Aydemir and Skuterud (2005) find that starting wages have declined for native workers as well. However, I will focus on – and estimate – the average wage gaps between immigrants and natives rather than their trajectories over time.

Finally, I close the model by combining the firm value functions and the free-entry condition to derive a job-creation condition:

$$q(\theta_i) \sum_{j} \phi_{ij} \frac{\pi_{ij} p_i - w_{ij}}{r + s_{ij}} = c_i.$$

$$(6)$$

The equilibrium of this model is defined by 10 equations: the capital first-order condition (1), the intermediate-good first-order conditions (2) and (3), the 4 wage conditions in (4), the government budget constraint (5), and the 2 job-creation conditions in (6). These equations define the equilibrium values for the 10 values $\{K, p_L, p_H, w_{LN}, w_{LI}, w_{HN}, w_{HI}, t, \theta_L, \theta_H\}$.

This completes the description of the theoretical model. As in Battisti et al. (2018), I will focus on a numerical analysis of the model, and particularly of the effect of immigration on equilibrium outcomes. The following section presents the calibration of the model in two different scenarios – with and without discrimination – and a comparison of the estimated fiscal impacts of immigration in those scenarios.

3 Calibration and Simulation

Given the large and increasing importance of immigration in the Canadian labour market, and the fact that Canada has been studied by numerous papers within the literature on the fiscal impact of immigration (Grubel and Grady, 2011; Javdani and Pendakur, 2014; Grady and Grubel, 2014; Zhang et al., 2020; Kapsalis, 2021; Montcho et al., 2022), I will calibrate my model to data for Canada, and use the calibrated model to simulate the fiscal effect of increases in the immigrant population. I will do this in each of two scenarios: one in which the model is calibrated under the assumption that wage differences between immigrants and natives are due to productivity differences, and one in which immigrants and natives are assumed to be of equal productivity and the wage differences are attributed to discrimination.

A number of the parameter values are common between the two calibrations, and their values are summarized in Table 1. In particular, I follow Battisti et al. (2018) in choosing several of the parameters to match the values that they used for all 20 countries in their study: I set $\varepsilon = 0.5$, $\rho = 0.5$, r = 0.004, and $\delta = 0.0061$. The UI benefit b_{ij} is modelled as being equal to a replacement rate ϱ multiplied by after-tax wages $w_{ij}(1-t)$, and the

tax-benefit calculator at OECD Benefits and Wages (2023) gives an effective replacement rate of $\varrho = 0.394$ for the default unemployed case in Canada in 2019.¹² The capital share is estimated at $\alpha = 0.345$, based on a share of labour compensation in GDP of 65.5% in 2019 in the Federal Reserve Bank of St. Louis FRED database (FRED Economic Data, 2023), and I assume that all capital used in the baseline equilibrium is domestically owned, so that $\bar{K} = K$ at baseline, though K – the capital actually used in production – is allowed to adjust in equilibrium if immigrant populations change. As in Battisti et al. (2018), I normalize $c_L = 0.5$ and $\pi_{iN} = 1.^{13}$

For the population shares, as well as a number of the moments used in the calibrations, I use data from the Canadian Labour Force Survey (LFS) in 2019 (Statistics Canada, 2019). Specifically, I combine the data from all 12 months of the LFS in 2019, and drop all individuals aged 65 and over as well as those who are not in the labour force, leaving me with a sample of 726053 observations.¹⁴ I identify high-skilled individuals as those with at least a bachelor's degree. To calculate the population shares, I normalize the size of the native population to 1, and then I can calculate $Q_{HN} = 0.270$, $Q_{LN} = 0.730$, $Q_{HI} = 0.157$, and $Q_{LI} = 0.186$.

The remaining parameters in each of the two calibrations are chosen to match a set of 11 moments which are presented in Table 2. Most of the moments are estimated in the data from the 2019 LFS, and the government expenditure as a percentage of GDP is taken from the data at IMF Datamapper (2022) for 2019; the two values for job durations from Battisti et al. (2018) were used for all 20 of their countries, including Canada.

In the first scenario, discrimination is absent from the model, which means that the worker bargaining power $\beta_{ij} = \beta$ does not vary by type, and in particular does not vary between natives and immigrants within a given skill type. To be precise, as in Battisti et al. (2018), β will be set equal to ε , with both equal to 0.5. I then have a set of 11

¹²The default case corresponds to a 40-year-old individual in a couple with 2 children (aged 4 and 6), unemployed for 2 months after working for 264 months over their career and with a previous wage equal to the average, and with a non-employed spouse.

¹³Note that setting $\pi_{LN} = \pi_{HN} = 1$ is without loss of generality since x can vary to determine the relative productivity of low- and high-skill workers.

¹⁴Of course, given the fact that the LFS surveys individuals over 6 consecutive months, many people appear in the data multiple times. This does not affect the representativeness of the overall sample, and I use the "standard final weight" (variable FINALWT) to weight all estimations.

Table 1: Common Parameters Taken from Available Data or Literature

Parameter	Description	Value	Source	
β (or β_N)	worker bargaining power	0.5	Battisti et al. (2018)	
arepsilon	matching elasticity	0.5	Battisti et al. (2018)	
ho	controls substitution elasticity	0.5	Battisti et al. (2018)	
r	monthly interest rate	0.004	Battisti et al. (2018)	
δ	monthly depreciation rate	0.0061	Battisti et al. (2018)	
ϱ	UI replacement rate	0.394	OECD Benefits and Wages	
α	capital share	0.345	FRED Database	
Q_{LN}	low-skilled natives	0.730	Labour Force Survey	
Q_{HN}	high-skilled natives	0.270	Labour Force Survey	
Q_{LI}	low-skilled immigrants	0.186	Labour Force Survey	
Q_{HI}	high-skilled immigrants	0.157	Labour Force Survey	
c_L	low-skilled vacancy cost	0.5	normalization	
$\pi_{iN} = 1$	native labour productivity	1	normalization	

Notes: The parameters in this table are taken from the sources listed in the final column, and are used in both calibrations of the model with and without discrimination, with the partial exception of β : in the scenario without discrimination, all $\beta_{ij} = \beta = 0.5$, whereas in the scenario with discrimination, $\beta_N = 0.5$ for natives, while the values for immigrants are chosen later to match moments in the data.

Table 2: Moments Matched in Calibration

Moment	Value	Source
average job durations, low-skilled (months)	29.4	Battisti et al. (2018)
average job durations, high-skilled (months)	52.6	Battisti et al. (2018)
native wage premium, low-skilled	1.085	Labour Force Survey
native wage premium, high-skilled	1.151	Labour Force Survey
skilled-unskilled wage ratio, native workers	1.449	Labour Force Survey
unemployment rate, low-skilled natives	0.065	Labour Force Survey
unemployment rate, low-skilled immigrants	0.065	Labour Force Survey
unemployment rate, high-skilled natives	0.034	Labour Force Survey
unemployment rate, high-skilled immigrants	0.054	Labour Force Survey
government expenditure as $\%$ of GDP	0.4065	IMF
real GDP per capita	1	normalization

Notes: The moments in this table are taken from the sources listed in the final column, and are used in both calibrations of the model with and without discrimination.

parameters given by $\{\xi, A, x, c_H, g, s_{ij}, \pi i I\}$ to calibrate, and the results of the calibration – the parameter values which cause the simulated model to match the moments in Table 2 – are presented in Table 3. Many of the parameter values are fairly close to the averages found by Battisti et al. (2018), though our model is somewhat different; unsurprisingly, job

separation rates are higher for the low-skilled, and somewhat for high-skilled immigrants, to explain their higher unemployment rates. The values of the 10 variables in equilibrium (and in the immigration scenarios to come) can be found in appendix A.

Table 3: Calibrated Parameter Values with No Discrimination

Parameter	Description	Value
${\xi}$	match efficiency parameter	0.818
A	total factor productivity	0.503
x	low-skill intermediate share	0.506
c_H	cost of high-skill vacancy	0.870
g	lump-sum transfer	0.295
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0341
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
π_{LI}	labour productivity, low-skilled immigrants	0.922
π_{HI}	labour productivity, high-skilled immigrants	0.874

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the absence of discrimination.

Before proceeding to policy simulations, I can also present the results of the second calibration scenario, in which the wage gaps between natives and immigrants are explained by discrimination rather than productivity differences. In this scenario, natives and immigrants of a given skill type have the same productivity, so all productivity parameters π_{ij} are set to 1, but immigrants face discrimination in the form of lower levels of bargaining power: the bargaining power parameters are now β_N for all natives, and β_{LI} and β_{HI} for immigrants. In other words, in this scenario, the entire wage gap between immigrants and natives is explained by discrimination, ¹⁵ and once again I set $\beta_N = \varepsilon = 0.5$. This specification for discrimination is a simple way of modelling the idea that immigrants are less able to bargain a fair wage than natives; however, a more traditional form of taste-based discrimination in

 $^{^{15}}$ Of course, this is not necessarily the maximum possible extent of discrimination: if immigrants are positively selected on unobservable characteristics such as determination or desire to work, immigrants' wages could actually be higher than native wages in the absence of discrimination and observable productivity differences.

 $^{^{16}}$ In additional results that are available upon request, I find that if β and β_N are smaller than 0.5 – so that even native workers are paid less than the efficient wage – the results in Table 5 are almost identical in the scenario without discrimination, whereas the simulation method gives slightly less positive effects of immigration in the scenario with discrimination. As a result, the effect of discrimination on the estimated fiscal impact of immigration is slightly smaller, but still very significant.

which employers receive disutility from hiring workers from a particular category (immigrants in this case) gives similar results, as demonstrated in section 4.5.

The parameter values that generate a match to the moments in Table 2 in the presence of discrimination are presented in Table 4. Most of the parameters are similar to their values from Table 3, with the exception of c_H , which is considerably larger to explain why firms don't create more high-skilled vacancies to take advantage of relatively low wages of high-skilled immigrants. The wage bargaining parameters for immigrants are quite low, as such values are required to explain the difference in wages with respect to natives in the absence of productivity differences. As before, the equilibrium values of the 10 variables can be found in appendix A.

Table 4: Calibrated Parameter Values with Discrimination

Parameter	Description	Value
ξ	match efficiency parameter	0.638
A	total factor productivity	0.494
x	low-skill intermediate share	0.502
c_H	cost of high-skill vacancy	2.177
g	lump-sum transfer	0.295
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0341
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
eta_{LI}	bargaining power, low-skilled immigrants	0.171
eta_{HI}	bargaining power, high-skilled immigrants	0.104

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the presence of discrimination.

Having calibrated the parameters of my model, I can now perform policy simulations in which I consider various scenarios of increased immigration, and calculate the impact of that immigration on the fiscal situation of the government. The 3 scenarios that I consider are (i) a 10% increase in Q_{LI} , (ii) a 10% increase in Q_{HI} , and (iii) a 10% increase in both Q_{LI} and Q_{HI} at the same time. I calculate the fiscal impact of each increase in immigrant numbers in two ways: first, a naive method that simply counts the taxes paid and benefits received by the new immigrants, and then a simulation method which allows for changes to all endogenous variables by solving for the new equilibrium after the increase in immigration,

but holding the value of t fixed. I report the effect per immigrant, dividing the total effect on the government budget by the change in the immigrant population.

Table 5 presents the results of these simulations for both calibration scenarios. In the absence of discrimination, we can see in panel A that the naive method gives quite similar results to the simulation method; the only difference between the two approaches is that the latter permits general equilibrium adjustments to wages and other variables, but this does not significantly impact the results in this version of the model. The results indicate that low-skilled immigrants impose a fiscal cost on the rest of the population: the naive method finds that they pay average taxes of 0.2488 and receive benefits of 0.3025, for a net cost of about 5.4% of average income, and the simulation method gives similar results. High-skilled immigrants, on the other hand, produce a fiscal surplus of around 4% of average income, due to higher tax payments (0.3439, versus benefits received of 0.3035). A simultaneous increase in the size of both immigrant populations produces a small fiscal burden of slightly over 1% of average income per immigrant.

Table 5: Fiscal Impact of Increased Immigration

	Scenario 1	Scenario 2	Scenario 3
Method	10% Increase in Q_{LI}	10% Increase in Q_{HI}	10% Increase in Both
	Panel A	: No Discrimination	
naive	-0.0537	0.0404	-0.0106
simulation	-0.0553	0.0374	-0.0120
	Panel	B: Discrimination	
naive	-0.0537	0.0404	-0.0106
simulation	-0.0269	0.0648	0.0161

Notes: This table presents the estimated fiscal impact of each new immigrant in each scenario.

To put these results into units that can be more easily interpreted, the average income of my Labour Force Survey sample is \$1042, for an annual income of \$52084 for an individual who works for 50 weeks per year, so my results suggest that, in the absence of discrimination, the average immigrant costs taxpayers about \$624 per year, while low-skilled immigrants' costs are larger at about \$2880 and high-skilled immigrants produce fiscal gains of about \$1946. In additional results that are available upon request, I also calculate the effect of

immigration on the welfare of each type of worker (now allowing t to adjust to balance the government budget), and I find that low-skilled immigration provides welfare gains to each type of worker except for the already-present low-skilled immigrant population, whereas high-skilled immigration helps the low-skilled and harms the high-skilled, and an increase in both types helps all but high-skilled immigrants. In each case, the impact on average welfare is positive.

In the scenario in which immigrant wage gaps are explained by discrimination, panel B presents the results; unsurprisingly, the results with the naive method are roughly identical to those without discrimination: the two versions of the model are calibrated to the same moments, including wages and unemployment rates, and so the calculations of the taxes paid and benefits received by different groups are extremely similar. However, the results from the simulation method tell a different story: the fiscal cost of a low-skilled immigrant is less than half as large as it was without discrimination, while the fiscal benefit of high-skilled immigrants is significantly larger at 6.5% instead of 3.7%. A simultaneous increase in the size of both immigrant populations now generates a fiscal benefit rather than a cost. In dollars, a low-skilled immigrant's fiscal cost is now \$1403 instead of \$2880, and the fiscal gain from a high-skilled immigrant rises from \$1946 to \$3373, while the average immigrant now benefits taxpayers by about \$839 instead of costing \$624.

The mechanism for this fiscal effect of discrimination against immigrants is fairly simple: discrimination against immigrants leads to an increased surplus for firms, who have the chance of hiring immigrants and making profits due to their lower wages, which leads to greater vacancy creation as the higher surplus raises the value of a vacancy above zero. Because of the zero-profit condition, intermediate-goods firms will continue creating vacancies up to the point at which the value drops to zero, meaning that their profits are unaffected in equilibrium, and final-good firms also make zero profits due to constant returns to scale. So the effect takes place through tax payments and benefit receipt of workers: increased vacancy creation means a higher probability of being employed, which means that workers pay more in taxes and receive UI benefits less often.

Accounting for wage discrimination against immigrants thus has the potential to significantly alter our conclusions about the fiscal impact of immigration; in particular, the overall effect of immigration would be estimated to be negative if we ignore discrimination, and is positive when discrimination is modelled. The Labour Force Survey suggests that the immigrant population of working age was 4.865 million in Canada in 2019, which suggests that the total fiscal burden of immigration would be \$3.04 billion if we don't account for discrimination, whereas accounting for discrimination leads to the conclusion that immigration produces a fiscal surplus of \$4.08 billion per year. Wage discrimination against immigrants clearly has a negative impact on the immigrants themselves, but my results suggest that it may generate a net fiscal benefit of \$7.12 billion per year to governments in Canada if it explains the wage gaps between native and immigrant workers.

As in the case of the model without discrimination, I can also calculate the effect of immigration on welfare, and the results (available upon request) show the same direction of effects, though with a more positive overall effect of immigration on welfare.

4 Alternative Models and Robustness Checks

In this section, I will extend the analysis of the previous section to consider several alternative modelling specifications. First, I consider a scenario in which part of government spending is on public goods, the cost of which does not increase with population size; then I model different tax rates applying to different groups, to account for progressive taxation. Subsequently, I present two important modifications to the main model: one in which final-goods firms vary by total factor productivity A and pay taxes on their profits, and one in which capital is taxed in a closed-economy setting. Finally, I model discrimination as classic taste-based discrimination on the part of employers. All simulation results can be found in Table 6; in each case, I show that the effect of discrimination on the fiscal impact of immigration remains significant, though the quantitative findings vary in each case.

4.1 Public Goods

In the main model in section 2, I assumed that the government only pays for the UI benefit b_{ij} and the lump-sum transfer g. However, part of government spending is on things like national defence and foreign affairs, among other things, which one might consider to be public goods that would not depend on the size of the population. In that case, immigration

can produce an added fiscal benefit by spreading the cost of those public goods over a larger population; or, put another way, the marginal fiscal cost of a new person is lower than the average cost. To model this, I assume that some fraction of g is not a lump-sum transfer, but rather a public good that each new immigrant receives without the government needing additional funding; and I use 10% as the public-good fraction of g, since it is in the middle of the plausible range of $\{4.3\%, 10\%, 15\%\}$ of total tax revenues considered by Javdani and Pendakur (2014).

The calibration of the model is unchanged, and the new simulations of the immigration experiments produce the results that can be found in Table 6. In the absence of discrimination, the results from the two methods once again agree, but on a fiscal impact of immigration that is much more positive than in section 3, because immigrants don't require as much additional government spending as was assumed there. In the presence of discrimination, the naive results are unchanged as before, but the simulation method results are even more positive: even low-skilled immigrants now produce a fiscal benefit. However, the overall effect of discrimination on the fiscal impact of immigration is very similar to the baseline model: the total fiscal surplus from immigration in Canada is estimated at \$7.26 billion without discrimination and \$14.38 billion with discrimination, so the effect of discrimination is estimated at \$7.12 billion, exactly as in section 3.

4.2 Different Tax Rates

In this subsection, I consider an alternative modification to the model: individuals of different types may face different income tax rates, due to their different income levels and the progressive nature of the income tax in Canada. Specifically, natives tend to receive higher incomes, and thus are likely to pay higher marginal tax rates – and when considering the effect of discrimination, which may shift income between individuals of different types, the marginal tax rate is the relevant value.

To account for this possibility, I apply the federal and provincial income taxes from 2019 to my Labour Force Survey data: I use weekly wages, multiplying by 50 on the assumption that people work for 50 weeks per year, and I apply the tax schedules accounting only for the personal exemption. I find that the marginal tax rate is lowest for low-skilled immi-

Table 6: Fiscal Impact of Increased Immigration (Robustness Analyses)

	Scenario 1	Scenario 2	Scenario 3
Method	10% Increase in Q_{LI}	10% Increase in Q_{HI}	10% Increase in Both
	Panel A: No D	iscrimination	
naive (public goods)	-0.0130	0.0810	0.0300
sim. (public goods)	-0.0146	0.0780	0.0287
naive (different taxes)	-0.0723	0.0459	-0.0182
sim. (different taxes)	-0.0685	0.0344	-0.0205
naive (Melitz)	-0.1003	-0.0266	-0.0666
sim. (Melitz)	-0.1170	-0.0511	-0.0865
naive (closed economy)	-0.0857	-0.0041	-0.0483
sim. (closed economy)	-0.0552	0.0372	-0.0120
	Panel B: Dis	crimination	
naive (public goods)	-0.0130	0.0810	0.0300
sim. (public goods)	0.0137	0.1054	0.0568
naive (different taxes)	-0.0718	0.0457	-0.0180
sim. (different taxes)	-0.0404	0.0613	0.0072
naive (Melitz)	-0.1018	-0.0269	-0.0676
sim. (Melitz)	-0.0957	-0.0304	-0.0654
naive (closed economy)	-0.0866	-0.0051	-0.0493
sim. (closed economy)	-0.0267	0.0679	0.0177
naive (taste-based)	-0.0693	0.0187	-0.0291
sim. (taste-based)	-0.0402	0.0741	0.0132

Notes: This table presents the estimated fiscal impact of each new immigrant in each scenario.

grants, while low-skilled natives pay a rate that is 2.3% higher on average, and high-skilled immigrants and natives pay rates that are 4.3% and 6.9% higher respectively. I assume, for simplicity, that the UI tax rate is constant across types and equal to the lowest rate. I then recalibrate the model with and without discrimination, with the results found in Tables 7 and 8; unsurprisingly, the parameter values change very little.

I then evaluate the fiscal impact of immigration as before, with results that can be found in Table 6. Given that the modification to the model involves applying lower tax rates to low-skilled immigrants, it is not surprising that the fiscal impact of low-skilled immigration is more negative now, while the impact of high-skilled immigration is roughly the same as before. But the impact of discrimination remains very similar to the baseline model: accounting for discrimination significantly lowers the fiscal cost of low-skilled immigrants,

Table 7: Calibrated Parameter Values with No Discrimination with Different Tax Rates

Parameter	Description	Value
${\xi}$	match efficiency parameter	0.813
A	total factor productivity	0.502
x	low-skill intermediate share	0.506
c_H	cost of high-skill vacancy	0.766
g	lump-sum transfer	0.294
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0340
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0251
π_{LI}	labour productivity, low-skilled immigrants	0.923
π_{HI}	labour productivity, high-skilled immigrants	0.875

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the absence of discrimination and allowing for different marginal tax rates for each type.

Table 8: Calibrated Parameter Values with Discrimination with Different Tax Rates

Parameter	Description	Value
${\xi}$	match efficiency parameter	0.638
A	total factor productivity	0.494
x	low-skill intermediate share	0.502
c_H	cost of high-skill vacancy	2.108
g	lump-sum transfer	0.295
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0340
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
eta_{LI}	bargaining power, low-skilled immigrants	0.180
β_{HI}	bargaining power, high-skilled immigrants	0.099

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the presence of discrimination and allowing for different marginal tax rates for each type.

raises the fiscal benefit of high-skilled immigrants, and reverses the sign of the average effect of immigration, with the total fiscal impact going from a cost of \$5.21 billion to a benefit of \$1.81 billion, or an impact of discrimination of \$7.02 billion.

4.3 Simplified Melitz Model

The model used so far allowed for indirect fiscal effects of immigration through the employment of other workers, as discussed in section 3: discrimination against immigrants generates a surplus that encourages vacancy creation, producing fiscal gains from higher income tax payments and lower UI spending. The assumptions of the model meant that this was the only possible channel for a fiscal impact of discrimination: intermediate-good firms face a zero-profit condition, and final-good firms make zero profits due to constant returns to scale.

In the current subsection, I will modify the model to allow for profits for firms, and thus for corporate taxation. Specifically, while intermediate-goods firms are unaltered, I assume that the productivity A of final-goods firms comes from a distribution G(A); while I do not model firm entry, this is essentially a simplified form of a model in the spirit of Melitz (2003) in which entrepreneurs enter the market and draw a random productivity parameter. As in such models, I assume that only firms with positive profits will operate, which will be all those firms with A above an endogenous zero-profit cutoff value \bar{A} , accounting for a fixed cost of production f. However, to avoid degenerate results for firm size and profits, I have to model decreasing returns to scale in production, and I also abstract from capital for simplicity; rather than receiving the returns from capital, the native workers all receive an equal share of the firms' profits after they are taxed at rate τ . In the calibration, I model τ as being in a fixed proportional relationship with t: in 2019, OECD.Stat (2023b) found that the average corporate tax rate in Canada was 26.62%, whereas the marginal income tax rate (the "total tax wedge"), averaged across four income levels, was 38.375% in OECD.Stat (2023a), so I model $\tau = \frac{0.2662}{0.38375}t$.

In this version of the model, the final output for a firm with a given A is now given by:

$$Y = A [xY_L^{\rho} + (1 - x)Y_H^{\rho}]^{\frac{\alpha}{\rho}}, \alpha \in (0, 1), \rho \in (0, 1)$$

where $\alpha < 1$ generates decreasing returns to scale. Aside from replacing rk_{ij} with a share of the profits in the workers' Bellman equations, nothing else in the basic structure of the model changes, but the market-clearing conditions in intermediate goods are a bit different: a profit-maximizing firm with productivity A will demand Y_L and Y_H according to $\frac{\partial Y}{\partial Y_i} = p_i$, which means:

$$A\alpha x Y_L^{\rho-1} \left[x Y_L^{\rho} + (1-x) Y_H^{\rho} \right]^{\frac{\alpha-\rho}{\rho}} = p_L$$
$$A\alpha (1-x) Y_H^{\rho-1} \left[x Y_L^{\rho} + (1-x) Y_H^{\rho} \right]^{\frac{\alpha-\rho}{\rho}} = p_H.$$

This means that the ratio of Y_H to Y_L is independent of A:

$$\frac{Y_H}{Y_L} = \left(\frac{(1-x)p_L}{xp_H}\right)^{\frac{1}{1-\rho}}$$

but firms with high A will be larger:

$$Y_{L} = \left(\frac{A\alpha x}{p_{L}}\right)^{\frac{1}{1-\rho}} \left[x + (1-x)\left(\frac{(1-x)p_{L}}{xp_{H}}\right)^{\frac{\rho}{1-\rho}}\right]^{\frac{\alpha-\rho}{\rho(1-\alpha)}}$$

$$Y_{H} = \left(\frac{A\alpha(1-x)}{p_{H}}\right)^{\frac{1}{1-\rho}} \left[x + (1-x)\left(\frac{(1-x)p_{L}}{xp_{H}}\right)^{\frac{\rho}{1-\rho}}\right]^{\frac{\alpha-\rho}{\rho(1-\alpha)}}.$$

Therefore, equations (2) and (3) are replaced by equations that define the prices p_L and p_H such that the integrals of values of Y_L and Y_H demanded by firms that produce are equal to the amounts of Y_L and Y_H produced by intermediate firms.

For the calibration, I no longer need to choose a value of δ , since there is no capital in the model, and r is now just the discount rate and still equal to 0.004. α is now the decreasing-returns-to-scale parameter rather than the capital share, and I arbitrarily set it to 0.75; otherwise, there is no change to the set of parameters from Table 1. Meanwhile, I still have the same set of 11 parameters to fit based on the moments, except that I need to replace A with parameters of the distribution G(A): I assume a Pareto distribution with a minimum value of x_m (and a maximum value of 10) and a shape parameter η , so that $G(A) = \frac{\eta x_m^{\eta}}{A^{\eta+1}}$. I also need to choose a value for f, the fixed cost of production, and so in the absence of discrimination the set of 13 parameters to calibrate is $\{\xi, x_m, f, \eta, x, c_H, g, s_{ij}, \pi iI\}$, whereas π_{iI} are replaced by β_{iI} in the presence of discrimination.

The 11 moments previously used in the calibration are used again here, along with 2 new moments: the percentage of workers in firms with at least 100 employees (where I treat workers who produce the intermediate goods used by a final firm as being employees of that final goods firm), which was 61% in 2019 according to Statistics Canada (2022); and the 10-year survival rate of new firms (which I model as the percentage of entrepreneurs in the distribution of A that actually produce), which was 45.3% during the period of 2001-2017 according to Innovation, Science and Economic Development Canada (2020).

The calibration results can be found in Tables 9 and 10, where we can see that the

parameters that existed in the baseline model are not dramatically different here.¹⁷ The fiscal impact of immigration can be found in Table 6, and the overall effect of immigration on the government's budget balance is much more negative now, because income taxes are lower and immigrants are assumed to have no share in the profits, so their fiscal contribution is smaller. There is also a larger difference between the naive and simulation methods, with the simulation method being even more negative in the absence of discrimination. However, the effect of discrimination remains similar when we compare the results from the simulation methods in the two cases:¹⁸ in each case, accounting for discrimination makes the effect of immigration significantly less negative. At an economy-wide level, the fiscal cost of immigration is \$21.93 billion in the absence of discrimination and \$16.57 billion in its presence, for an effect of discrimination of \$5.35 billion.

Table 9: Calibrated Parameter Values with No Discrimination in Melitz Model

Parameter	Description	Value
ξ	match efficiency parameter	0.701
x_m	minimum value of A	0.514
η	shape parameter of $G(A)$	2.281
f	fixed cost of production	0.0062
x	low-skill intermediate share	0.506
c_H	cost of high-skill vacancy	0.972
g	lump-sum transfer	0.291
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0341
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
π_{LI}	labour productivity, low-skilled immigrants	0.922
π_{HI}	labour productivity, high-skilled immigrants	0.875

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2, as well as the percentage of workers in firms with at least 100 employees and the 10-year survival rate of new firms, in the absence of discrimination in the Melitz-type model.

 $^{^{17}}$ Unlike the other models, it was not possible to perfectly match the moments in these calibrations, due to the discretization of the distribution of A and the greater complexity of the model; but minimizing the sum of squared deviations between moments in the data and the simulation, the fit remained very close, corresponding to an average absolute deviation of 0.0015 or less.

¹⁸The naive method does not properly account for discrimination, so comparing it to the simulation method in the presence of discrimination is not a valid comparison.

Table 10: Calibrated Parameter Values with Discrimination in Melitz Model

Parameter	Description	Value
ξ	match efficiency parameter	0.553
x_m	minimum value of A	0.515
η	shape parameter of $G(A)$	2.316
f	fixed cost of production	0.0065
x	low-skill intermediate share	0.502
c_H	cost of high-skill vacancy	1.976
g	lump-sum transfer	0.292
s_{LN}	monthly job separation rate, low-skilled natives	0.0341
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0338
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
eta_{LI}	bargaining power, low-skilled immigrants	0.197
eta_{HI}	bargaining power, high-skilled immigrants	0.127

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2, as well as the percentage of workers in firms with at least 100 employees and the 10-year survival rate of new firms, in the presence of discrimination in the Melitz-type model.

4.4 Closed-Economy Model

I next return to the original model with its single value of A and constant-returns-to-scale production including capital, but I now assume a closed economy in which capital is supplied by natives and demanded by firms according to $r + \delta = \alpha A K^{\alpha-1} Z^{1-\alpha}$. If natives face a discount rate of r, then in a dynamic model they will supply capital through their savings up to the point at which its return is r; as a result, r will remain fixed at its initial value of 0.004, but now domestically-held capital \bar{K} adjusts one-for-one with K. I model the steady-state, so I do not need to account for the gradual accumulation of capital over time.

As a result, all that changes relative to the baseline model is the native utility function, as the amount of capital held in steady-state varies rather than remaining constant, and I assume that that capital is now taxed at the same rate that firm profits were in the Melitz-type model in section 4.3. I calibrate the model to the same moments from Table 2, using the values from Table 1 for the other parameters, and the results of the calibration can be found in Tables 11 and 12; the parameter values are very similar to those from the baseline model.

The immigration scenario simulations take the same form as before, and the results can

Table 11: Calibrated Parameter Values with No Discrimination in Closed-Economy Model

Parameter	Description	Value
ξ	match efficiency parameter	0.776
A	total factor productivity	0.502
x	low-skill intermediate share	0.506
c_H	cost of high-skill vacancy	0.870
g	lump-sum transfer	0.294
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0341
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0252
π_{LI}	labour productivity, low-skilled immigrants	0.922
π_{HI}	labour productivity, high-skilled immigrants	0.875

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the absence of discrimination in the closed-economy model.

Table 12: Calibrated Parameter Values with Discrimination in Closed-Economy Model

	D : /:	T 7 1
Parameter	Description	Value
ξ	match efficiency parameter	0.617
A	total factor productivity	0.494
x	low-skill intermediate share	0.502
c_H	cost of high-skill vacancy	2.089
g	lump-sum transfer	0.294
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0342
s_{HN}	monthly job separation rate, high-skilled natives	0.0155
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0251
eta_{LI}	bargaining power, low-skilled immigrants	0.184
eta_{HI}	bargaining power, high-skilled immigrants	0.114

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the presence of discrimination in the closed-economy model.

be found in Table 6. As in the Melitz model, the fiscal impact tends to be more negative here, and for a similar reason: immigrants are assumed to own no capital. However, in the no-discrimination case, the fiscal impact is more positive in the simulation method, in which immigrants do implicitly contribute to capital tax revenues, and indeed those results are very similar to those from the baseline model. In the case with discrimination, the results of the simulation method are more positive still, again much like in the baseline model, and

at an economy-wide level discrimination raises the overall fiscal impact of immigration from a cost of \$3.04 billion to a benefit of \$4.48 billion. The results of this model are thus very similar to those from the baseline model.

4.5 Alternative Form of Discrimination

Finally, I consider one additional model to test the robustness of my results: I model discrimination as classic taste-based discrimination on the part of employers, rather than in the form of a lower β_{ij} parameter for immigrants. In this setting, of course, the results of the model without discrimination are unchanged, so I only need to consider the second calibration scenario with discrimination

The main change to the baseline model is that I now introduce a new parameter d_{ij} representing the disutility to the employer (an intermediate-good firm) from hiring a worker of type $\{i, j\}$; I normalize $d_{iN} = 0$ and interpret d_{iI} as the discrimination against immigrants of skill level i. Since it is no longer clear that firms will benefit from hiring immigrants – since they pay a disutility cost from doing so – profits may increase in equilibrium and there may not be an increase in vacancy creation when immigrant numbers increase, and so I need to allow the income of firms to be subject to taxation, which I suppose takes place at the same rate as worker income. As a result, while the Bellman equation for the value of an open vacancy is unchanged, the Bellman equation for the value of a filled job is now given by:

$$rJ_{ij}^{F} = (1-t)(\pi_{ij}p_i - w_{ij}) - d_{ij} - s_{ij} \left[J_{ij}^{F} - J_{i}^{V} \right]$$

The free-entry condition still means that $J_i^V = 0$ in equilibrium, but now that no longer means that average profits are zero, because of the disutility term d_{ij} which is not part of the profits of the firm, but rather a disutility to the firm's owner.

The rest of the model is unchanged from the baseline setting, except that the job-creation condition changes in parallel with the value function for a filled job, and government revenue is now equal to $t \sum_i \sum_j \pi_{ij} p_i E_{ij}$ instead of $t \sum_i \sum_j w_{ij} E_{ij}$. In the calibration, I set all $\beta_{ij} = \beta = 0.5$, while all the productivity parameters π_{ij} are set to 1 as in the usual discrimination setting; the d_i parameters are now set to explain the immigrant wage gaps. The results of the calibration can be found in Table 13, where the main differences from the baseline model

are a somewhat higher value of ξ and a significantly lower value of c_H .

Table 13: Calibrated Parameter Values with Taste-Based Discrimination

Parameter	Description	Value
ξ	match efficiency parameter	0.830
A	total factor productivity	0.494
x	low-skill intermediate share	0.504
c_H	cost of high-skill vacancy	0.865
g	lump-sum transfer	0.295
s_{LN}	monthly job separation rate, low-skilled natives	0.0340
s_{LI}	monthly job separation rate, low-skilled immigrants	0.0341
s_{HN}	monthly job separation rate, high-skilled natives	0.0154
s_{HI}	monthly job separation rate, high-skilled immigrants	0.0253
d_L	discrimination term, low-skilled immigrants	0.0257
d_H	discrimination term, high-skilled immigrants	0.0565

Notes: The parameters in this table are the values that cause the simulated model to match the moments listed in Table 2 in the presence of taste-based discrimination.

The immigration scenario simulations can be found at the bottom of Table 6, and are generally similar to those from Table 5: the fiscal impact of low-skilled immigrants is more negative, and the impact of high-skilled immigrants is more positive, but the average effect is quite similar. The aggregate fiscal benefit from immigration is now estimated at \$3.34 billion, or about \$6.38 billion more than in the case without discrimination. The mechanism of the effect is different from the baseline model, relying largely on increased revenues from employers (whose profits increase with immigration while they experience greater disutility from hiring immigrants), but the overall impact is very similar.

5 Conclusion

This paper has addressed the question of wage discrimination against immigrants, and how it might affect our conclusions about the fiscal impact of immigration. I have demonstrated that, in the presence of such discrimination, the fiscal benefit of immigration to a destination country could be much more positive than existing studies have found, because when immigrants are paid less than their marginal product, someone else is receiving that income and paying taxes on it. I have presented a search-and-matching model in the style of Battisti et al. (2018), which I calibrate to data from Canada, and I show that the impact of

discrimination on the fiscal impact of immigration can be significantly positive, perhaps on the order of \$7 billion per year. These results are robust to a variety of modifications and robustness checks.

This analysis is based on a simple model that abstracts from differences in fiscal impact over the life-cycle, as well as certain complexities of the tax and transfer system, among other things; a completely convincing estimate of the true fiscal impact of immigration in Canada would require a more complete model. But the current analysis has shown that discrimination could have a significant quantitative impact on the conclusion of the analysis of any realistic model, and gives reason to believe that the fiscal impact of immigration on destination countries may be more positive than had previously been believed.

References

- AMIOR, M. AND A. MANNING (2022): "Monopsony and the Wage Effects of Migration," Unpublished Paper.
- AYDEMIR, A. AND M. SKUTERUD (2005): "Explaining the Deteriorating Entry Earnings of Canada's Immigrant Cohorts," Canadian Journal of Economics, 38, 641–672.
- Bartolucci, C. (2014): "Understanding the Native-Immigrant Wage Gap using Matched Employer-Employee Data: Evidence from Germany," *Industrial and Labor Relations Review*, 67, 1166–1202.
- Battisti, M., G. Felbermayr, G. Peri, and P. Poutvaara (2018): "Immigration, Search and Redistribution: A Quantitative Assessment of Native Welfare," *Journal of the European Economic Association*, 16, 1137–1188.
- Bødker, S., R. H. Jacobsen, and J. R. Skaksen (2012): "Fiscal Costs and Benefits of High Skilled Immigration to a Generous Welfare State," Discussion Paper No. 2013-06, Norface Migration.
- BORJAS, G. J. (1995): "The Economic Benefits from Immigration," *Journal of Economic Perspectives*, 9, 3–22.
- BOUBTANE, E., J.-C. DUMONT, AND C. RAULT (2014): "Immigration and Economic Growth in the OECD Countries 1986-2006," Oxford Economic Papers, 68, 340–360.
- Bratsberg, B., O. Raaum, and K. Røed (2014): "Immigrants, Labour Market Performance and Social Insurance," *Economic Journal*, 124, F644–F683.
- Busetta, G., M. G. Campolo, and D. Panarello (2018): "Immigrants and Italian Labor Market: Statistical or Taste-Based Discrimination," *Genus*, 74.

- Chassamboulli, A. and T. Palivos (2014): "A Search-Equilibrium Approach to the Effects of Immigration on Labor Market Outcomes," *International Economic Review*, 55, 111–129.
- Chassamboulli, A. and G. Peri (2020a): "The Economic Effect of Immigration Policies: Analyzing and Simulating the U.S. Case," *Journal of Economic Dynamics & Control*, 114, 103898.
- CLEMENS, M. (2021): "The Fiscal Effect of Immigration: Reducing Bias in Influential Estimates," Working Paper No. 9464, CESifo.
- Colas, M. and D. Sachs (2022): "The Indirect Fiscal Benefits of Low-Skilled Immigration," *American Economic Journal: Economic Policy*, forthcoming.
- Dustmann, C. and T. Frattini (2014): "The Fiscal Effects of Immigration to the UK," *Economic Journal*, 124, F593–F643.
- EDO, A., L. RAGOT, H. RAPOPORT, S. SARDOSCHAU, A. STEINMAYR, AND A. SWEET-MAN (2020): "An Introduction to the Economics of Immigration in OECD Countries," *Canadian Journal of Economics*, 53, 1365–1403.
- ESSES, V. M., C. BENNETT-ABUAYYASH, AND N. LAPSHINA (2014): "How Discrimination Against Ethnic and Religious Minorities Contributes to the Underutilization of Immigrants' Skills," *Policy Insights from the Behavioral and Brain Sciences*, 1, 55–62.
- FAYS, V., B. MAHY, F. RYCX, AND M. VOLRAL (2021): "Wage Discrimination based on the Country of Birth: Do Tenure and Product Market Competition Matter?" *Applied Economics*, 53, 1551–1571.
- FERRER, A., D. A. GREEN, AND W. C. RIDDELL (2006): "The Effect of Literacy on Immigrant Earnings," *Journal of Human Resources*, 41, 380–410.
- FORTIN, N., T. LEMIEUX, AND J. TORRES (2016): "Foreign Human Capital and the Earnings Gap Between Immigrants and Canadian-Born Workers," *Labour Economics*, 41, 104–119.
- FRED ECONOMIC DATA (2023): "Share of Labour Compensation in GDP at Current National Prices for Canada," https://fred.stlouisfed.org/series/LABSHPCAA156NRUG, Federal Reserve Bank of St. Louis, viewed on January 31, 2023.
- GRADY, P. AND H. GRUBEL (2014): "Immigration and the Welfare State Revisited: Fiscal Transfers to Immigrants in Canada in 2014," https://www.fraserinstitute.org/sites/default/files/immigration-and-the-welfare-state-revisited.pdf, Fraser Institute.

- Green, D. A. and C. Worswick (2012): "Immigrant Earnings Profiles in the Presence of Human Capital Investment: Measuring Cohort and Macro Effects," *Labour Economics*, 19, 241–259.
- GRUBEL, H. AND P. GRADY (2011): "Immigration and the Canadian Welfare State 2011," https://www.researchgate.net/publication/228280590_Immigration_and_the_Canadian_Welfare_State_2011, Fraser Institute.
- HANSEN, M. F., M. L. SCHULTZ-NIELSEN, AND T. TRANAES (2017): "The Fiscal Impact of Immigration to Welfare States of the Scandinavian Type," *Journal of Population Economics*, 30, 925–952.
- Hennessey, G. and J. Hagen-Zanker (2020): "The Fiscal Impact of Immigration: A Review of the Evidence," Working Paper 573, ODI.
- HIRSCH, B. AND E. J. JAHN (2015): "Is There Monopsonistic Discrimination against Immigrants?" *Industrial and Labor Relations Review*, 68, 501–528.
- IMF DATAMAPPER (2022): "Government Expenditure, Percent of GDP," https://www.imf.org/external/datamapper/exp@FPP/, International Monetary Fund, viewed on February 1, 2023.
- INNOVATION, SCIENCE ECONOMIC Canada AND Development (2020): "Kev Small Business Statistics 2020," https://ised-isde.canada.ca/ site/sme-research-statistics/en/key-small-business-statistics/ key-small-business-statistics-2020, Small Business Branch, Research and Analysis Directorate.
- JAVDANI, M. AND K. PENDAKUR (2014): "Fiscal Effects of Immigrants in Canada," *Journal of International Migration and Integration*, 15, 777–797.
- Kampelmann, S. and F. Rycx (2016): "Wage Discrimination against Immigrants: Measurement with Firm-Level Productivity Data," *IZA Journal of Migration*, 5, 15.
- Kapsalis, C. (2021): "Fiscal Impact of Recent Immigrants to Canada," Canadian Public Policy, 47, 170–179.
- KERR, S. P. AND W. R. KERR (2011): "Economic Impacts of Immigration: A Survey," Working Paper No. 16736, NBER.
- LI, Q. AND A. SWEETMAN (2014): "The Quality of Immigrant Source Country Educational Outcomes: Do They Matter in the Receiving Country?" Labour Economics, 26, 81–93.
- Melitz, M. J. (2003): "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity," *Econometrica*, 71, 1695–1725.
- MITARITONNA, C., G. OREFICE, AND G. PERI (2017): "Immigrants and Firms' Outcomes: Evidence from France," European Economic Review, 96, 62–82.

- Montcho, G., J. Navaux, M. Merette, and Y. Carriere (2022): "Comparing Public Transfers between Immigrants and Natives in Canada," Unpublished Paper.
- NATIONAL ACADEMIES OF SCIENCES, ENGINEERING, AND MEDICINE (2017): "Past and Future Fiscal Impacts of Immigrants on the Nation," in *The Economic and Fiscal Consequences of Immigration*, ed. by F. D. Blau and C. Mackie, National Academies Press, 277–380.
- OECD BENEFITS AND WAGES (2023): "Tax-Benefit Web Calculator," https://www.oecd.org/els/soc/benefits-and-wages/tax-benefit-web-calculator/#d.en.500997, OECD, viewed on January 31, 2023.
- OECD.STAT (2023a): "Table I.4. Marginal Personal Income tax and Social Security Contribution Rates on Gross Labour Income," https://stats.oecd.org/index.aspx? DataSetCode=Table_I4, OECD, viewed on February 2, 2023.
- OREOPOULOS, P. (2011): "Why Do Skilled Immigrants Struggle in the Labor Market? A Field Experiment with Thirteen Thousand Resumes," *American Economic Journal: Economic Policy*, 3, 148–171.
- Peri, G. (2012): "The Effect of Immigration on Productivity: Evidence from U.S. States," Review of Economics and Statistics, 94, 348–358.
- Preston, I. (2014): "The Effect of Immigration on Public Finances," *Economic Journal*, 124, F569–F592.
- ROWTHORN, R. (2008): "The Fiscal Impact of Immigration on the Advanced Economies," Oxford Review of Economic Policy, 24, 560–580.
- Ruist, J. (2015): "The Fiscal Cost of Refugee Immigration: The Example of Sweden," *Population and Development Review*, 41, 567–581.
- STATISTICS CANADA (2019): "Labour Force Survey, 2019," Abacus Data Network, V1, UNF:6:YjyJLAAEwC7/1Orys54Mbw== [fileUNF]. https://hdl.handle.net/11272.1/AB2/ATGWRX.
- VARGAS-SILVA, C. (2015): "Chapter 16 The Fiscal Impact of Immigrants: Taxes and Benefits," in *Handbook of the Economics of International Migration*, 845–875.
- WARMAN, C., A. SWEETMAN, AND G. GOLDMANN (2015): "The Portability of New Immigrants' Human Capital: Language, Education, and Occupational Skills," *Canadian Public Policy*, 41, S64–S79.

- WARMAN, C., M. D. WEBB, AND C. WORSWICK (2019): "Immigrant Category of Admission and the Earnings of Adults and Children: How Far Does the Apple Fall?" *Journal of Population Economics*, 32, 53–112.
- WARMAN, C. AND C. WORSWICK (2015): "Technological Change, Occupational Tasks and Declining Immigrant Outcomes: Implications for Earnings and Income Inequality in Canada," *Canadian Journal of Economics*, 48, 736–772.
- YAO, L., J. B. BOLEN, AND C. R. WILLIAMSON (2022): "Are Economic Arguments against Immigration Missing the Boat? The Fiscal Effects of the Mariel Boatlift," *Southern Economic Journal*, 89, 305–325.
- ZHANG, H., J. ZHONG, AND C. DE CHARDON (2020): "Immigrants' Net Direct Fiscal Contribution: How Does it Change over their Lifetime?" Canadian Journal of Economics, 53, 1642–1662.

A Equilibrium Values of Variables

In Table 14 below, I present the values of the 10 model variables in equilibrium, in the two baseline scenarios (with and without discrimination) and in each of the increased-immigration scenarios. The values of t do not change across scenarios (for a given calibration) by construction, because the immigration scenarios involve calculating the effect of immigration on the government budget balance holding t fixed.

Discrimination No Discrimination Baseline Scenario 1 Scenario 3 Baseline Scenario 1 Scenario 2 Scenario 3 Variable Scenario 2 K45.87546.39546.48247.00545.87546.45546.567 47.1510.6270.6240.6310.6280.609 0.606 0.614 0.611 p_L 0.9000.9050.8960.8800.8910.8750.8650.871 p_H 0.4710.4710.4710.4710.4850.4850.4850.485t θ_L 0.3580.3560.3610.3590.5890.607 0.5940.612 0.2910.2920.2870.2880.4810.4910.494 θ_H 0.4770.5930.6000.5980.6130.6100.6170.6140.596 w_{LN} 0.5650.5620.5680.5660.5490.5470.5530.552 w_{LI} 0.8930.8540.8880.8790.8840.8630.8690.859 w_{HN} 0.7710.7760.7640.7680.750 0.7550.7430.748 w_{HI}

Table 14: Equilibrium Values of Variables

Notes: The values in this table are the equilibrium values of the 10 variables in each possible scenario; t does not change between the baseline and immigration scenarios by construction, since the immigration scenario simulations estimate the fiscal impact holding t constant.